



A Review on Effective Use of Daylight Harvesting Using Intelligent Lighting Control Systems for Sustainable Office Buildings in India

Gnana Swathika Odiyur Vathanam ¹, Karthikeyan Kalyanasundaram ², Rajvikram Madurai Elavarasan ^{3,*}, Shabir Hussain Khahro ⁴, Umashankar Subramaniam ^{5,*}, Rishi Pugazhendhi ⁶, Mehana Ramesh ¹ and Rishi Murugesan Gopalakrishnan ¹

- ¹ School of Electrical Engineering, VIT Chennai, Chennai 600127, India; gnanaswathika.ov@vit.ac.in (G.S.O.V.); mehana.r2017@vitstudent.ac.in (M.R.); rishim.g2017@vitstudent.ac.in (R.M.G.)
- ² Larsen and Toubro Limited, Chennai 603111, India; kskk@lntecc.com
- ³ Clean and Resilient Energy Systems (CARES) Laboratory, Texas A&M University, Galveston, TX 77553, USA
- ⁴ Department of Engineering Management, College of Engineering, Prince Sultan University, Riyadh 11586, Saudi Arabia; shkhahro@psu.edu.sa
 - ⁵ Department of Communications and Networks, Renewable Energy Laboratory, College of Engineering, Prince Sultan University, Riyadh 11586, Saudi Arabia
 - Department of Mechanical Engineering, Sri Venkateswara College of Engineering, Chennai 602117, India; rishi2000.p@gmail.com
 - * Correspondence: rajvikram787@gmail.com (R.M.E.); usubramaniam@psu.edu.sa (U.S.)

Abstract: Lighting is a fundamental requirement of our daily life. A lot of research and development is carried out in the field of daylight harvesting, which is the need of the hour. One of the most desirable attributes of daylight harvesting is that daylight is available universally and it is a very clean and cost-efficient form of energy. By using the various methods of daylight harvesting, it is possible to attain the global Sustainable Development Goals. Daylight harvesting in the most fundamental sense is the lighting strategy control of the artificial light in an interior space where daylight is also present so that the required illumination level is achieved. This way, a lot of energy can be saved. Recently, in addition to energy efficiency, other factors such as cost-efficiency, user requirements such as uniform illuminance, and different levels of illuminance at different points are being considered. To simulate the actual daylight contribution for an office building in urban Chennai, India before construction, ECO TECH software is used by providing the inputs such as building orientation, and reflectance's values of the ceiling, wall, and floor to analyze the overall percentage of daylight penetration available versus the percentage prescribed in the Indian Green Building Council to obtain the credit points. Thus, the impact of architectural design on daylight harvesting and daylight predictive technology has experimented with office building in Chennai, India. This article will give an insight into the current trends in daylight harvesting technology and intends to provide a deeper understanding and spark a research interest in this widely potential field.

Keywords: daylight harvesting; sustainable building; smart lighting

1. Introduction

Energy is central to almost every significant challenge and opportunity that the world faces today. As the world's population continues to rise, so will the energy consumption. Nevertheless, a fossil-fuel-based economy prevails as a prime driver of climate change. Achieving affordable and clean energy is expected to galvanize efforts to reach the Paris climate change agreement and also to expedite the progress in Sustainable Development Goals [1]. Augmenting the clean energy penetration is utmost for which research on investigating the potential renewable energy resource in a country [2–5], policy assessment [6] and developing an optimized scenario with various hybrid clean energy technologies is



Citation: Odiyur Vathanam, G.S.; Kalyanasundaram, K.; Elavarasan, R.M.; Hussain Khahro, S.; Subramaniam, U.; Pugazhendhi, R.; Ramesh, M.; Gopalakrishnan, R.M. A Review on Effective Use of Daylight Harvesting Using Intelligent Lighting Control Systems for Sustainable Office Buildings in India. *Sustainability* 2021, *13*, 4973. https://doi.org/10.3390/ su13094973

Academic Editors: Tullio De Rubeis and Luca Evangelisti

Received: 16 March 2021 Accepted: 22 April 2021 Published: 29 April 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). pivotal [7]. A study suggests that the lack of energy efficient infrastructure and resiliency to climate change as the major weakness in progressing towards sustainability [8]. Thus, efficient use of energy as well as minimizing the energy consumption is desideratum.

The annual electricity consumed in commercial buildings and households is about 20% to 40% per year [9]. The Lighting energy makes up a major portion of the total energy consumed. However, the amount of daylight available within a building can vary throughout the day since the motion of the sun from sunrise to sunset is also based on dynamic weather conditions. This energy can be saved by using different potential solutions. The costs and energy consumed to provide lighting for building occupants can be reduced phenomenally by implementing suitable daylight harvesting techniques. Since daylight is available in every part of the world and the energy produced by daylight harvesting is very clean and affordable, it is destined to be a promising solution by 2030. Cities predominantly require a cost-efficient energy system as the energy consumption is very high in cities. Daylight harvesting systems serves as a beneficial solution for the environment, to promote energy efficiency and to attain technological progress by 2030.

In [10], several lighting control methods and algorithms used in daylight harvesting are discussed. In recent decades, lighting controls have been broadly contemplated considering different variables such as light fixture design, framework topology, and occupancy satisfaction [10].

Lighting control systems are usually used to set the optimum brightness, select target sensors, and control light fixtures to be ON or OFF. However, in recent years a very significant feature in lighting systems automatically sets the light fixtures in accordance with the daylight lighting so that the total brightness in an area is consistent with the uniform distribution of light [11].

The general idea is that a daylight harvesting system includes a daylight sensor that distinguishes the natural and artificial illuminance in a workspace. Identifying the differences in illuminance allows the values of the contributions to be combined in several ways so that the desired total illuminance can be obtained [9]. Recently, lighting control techniques dependent on inhabitance and daylight illuminance have been reliably used to reduce electric energy consumption. The adoption of LEDs and the combination of data and communication technologies empower lighting control systems to be smarter [10].

For Closed-loop controllers, a light sensor is used to calculate the internal illuminance so that the controller will control the artificial light fixtures in accordance with the available daylight illuminance. Usually, these sensors are present closer to the occupant so that the controller can alter the lighting to increase the visual comfort of the occupant. Figure 1 shows a photosensor present on the desk [12]. Figure 2 shows a ceiling-mounted photosensor present in a workspace.



Figure 1. Photosensor on Desk.



Figure 2. Ceiling-Mounted Photosensor in Workspace.

Apart from the controller design, various control algorithms are used in daylight harvesting [12]. Daylight harvesting is an age-old concept and its effect can still be seen embedded in the architecture of ancient civilizations. The daylight harvesting systems include daylight prediction systems due to dynamic sky conditions. The estimation of the daylight metrics aids in increasing the efficiency of daylight harvesting. Various prediction models developed using time series prediction, nntool, nftool, and neural networks are discussed. Apart from controllers and algorithms, daylight harvesting can also be used using building geometrics. Furthermore, this paper discusses the various metrics used to obtain uniform and desired interior illumination using parameters such as Windowto-floor ratio, daylight factor, daylight autonomy, etc., by using building simulators.

The integration of daylight in buildings has significant potential to reduce energy use. However, the luminous environment can result in issues such as glare, headaches, eye strain if daylight designs are not well constructed. This thesis presents results from a broadscale analysis involving data analysis and survey of three large commercial buildings using multiple shading methods such as automatic blinds, electrochromic glazing and roller shades. Physical data were obtained in order to assess the glare and sunshine conditions and a survey was sent to employees who were working with the same company to evaluate standard of visual ease and comfort in the workplace. The following are the key outcomes from [13],

- Employees who were more comfortable with their proximity to sunshine were more likely to have higher perceived efficiency levels and higher happiness levels.
- In buildings 1 and 2, the satisfaction of the occupants with their office ranged substantially depending on their distance to the doors, and those nearest to the perimeter of the facility were more comfortable than those who sat farther away from the walls.

Daylighting design is a complex topic, and an integrated design process is necessary for a well-lit space to be designed carefully [13].

Section 2 discusses about the concepts of daylight harvesting using controllers and algorithms. Section 3 discusses in detail about techniques for daylight prediction. Section 4 further elaborates about the influence of building geometrics in daylight harvesting techniques. Section 5 discusses a case study of where ECO TECH software is used to analyze the overall percentage of daylight penetration available verses the percentage prescribed in the Indian Green Building Council to obtain the credit points. Thus, the impact of architectural design on daylight harvesting and daylight predictive technology has experimented. Section 6 discusses the conclusions and future scope of work for daylight saving concepts.

2. Algorithms and Control Systems for Daylight Harvesting

The growing interest in energy and cost-efficient systems without compromising on user comfort has led to a boom in the research and development of daylight harvesting systems. Over the years, a lot of algorithms and control mechanism for the regulation of artificial lighting based on the availability of daylight is developed. One such approach is the use of a genetic algorithm as an optimizing tool for controlling the artificial illuminance. This approach is based on the fundamental principle of daylight harvesting, i.e., the required artificial illuminance equals the total illuminance required minus the available daylight illuminance. In this approach [14–16], the illuminance of the required lighting systems is represented by (1),

$$E_R = E_D - DF_m \cdot E_O \tag{1}$$

(1) can be further written as (2)

$$E_R = B \cdot Y = E_D - DF \cdot E_O \tag{2}$$

The minimization of Z is done using the Genetic Algorithm, which is used as a tool to develop an operation schedule for the light fixtures which are further controlled by dimming and ON/OFF control of lighting strategy. This system is used in public premises to minimize the usage of lighting energy and maximize the use of natural light [14,15].

Another important factor that is widely considered in daylight harvesting systems is providing desired illumination levels at the user-defined points. A self-tuning multivariable controller is being used for this purpose. It is a closed-loop controller that guarantees the stability of the LED lighting system [16].

The use of Fuzzy logic controllers has also increased over the years. This is because they are considered to be the most appropriate controllers for lighting systems owing to their generalization of lighting levels settings, simplicity, and auto-control [17]. Based on the presence of the users, daylighting levels, and comfort to the user, a controller using fuzzy logic is designed to vary the lighting. This way the lighting energy which contributes to a significant portion of building energy is also saved and user comfort is also not compromised.

The DALI protocol is used as a communication medium between the LED luminaires and controllers. The DALI protocol is used, as it is an intelligent protocol that is developed especially for lighting. Moreover, every LED light fixture is fitted with a constant-current DALI driver that obtains signals and controls the light output [18]. Figure 3 represents the working of a smart LED lighting system with a fuzzy logic controller.



Figure 3. Block Diagram of Smart Lighting System.

Adding on, it is found that a non-adaptive fuzzy controller, when tuned successfully, fails to achieve uniform illuminance when an ON/OFF failure of the bulb occurs. This problem is overcome by the use of a fully adaptive controller technique. This adaptive fuzzy control logic is also known as FMRLC [19].

Another method that is used to reduce the electrical energy requirement of the LED luminaires is Safety Extra Low Voltage wiring systems which can also be used to control them. This is achieved by using a Low-Latency communication Network, centralized 48 Volt DC to control LED luminaires, which is capable of gathering real-time data. DC power and control systems are also implemented using sophisticated monitoring methods that are capable of providing daylight harvesting from sole luminaires, delicate dimming, temperature sensing, and occupancy control. The dimming of the LEDs is controlled by PWM controllers which deliver a variable current ranging from 0 to an acceptable current

value of the light fixture. The LEDs are turned ON and OFF at high values of frequency which creates a square wave signal and the brightness of the LED is determined by the duty cycle of this signal [20].

A Neural network-based control strategy is also being used frequently these days. It is because it enables the design of a lighting control system where the user controls the lighting system without having any knowledge of the system parameters [21]. On the other hand, daylight as a disturbance involves using a controller whose main intent is enhancing the efficiency of lighting energy via feedback control of artificial illumination devices. The lighting control algorithm is specified as a MIMO system where power loaded to the luminaires is considered to be the input and the light level at the target points is the derived output [22]. Another approach commonly used for NZEB is to model a luminance system via ANN and to use it for controller design using the IMC principle [23].

Photodetectors are often used to continuously calculate daylight illumination and use it for the neural network as a bias value. It is ensured by this method that each region receives the desired level of power consumption, uniform lighting, and the required level of lighting [24]. An open-loop control system based on sensed values of daylight and information for adjustment obtains the fading rates of the lighting fixture for adjusting daylight and occupancy shift. Besides, the efficiency of the daylight sensing lighting control system is found to be more robust as compared to the photodetector-based system in the presence of changes of reflectivity in the environment. Figures 4 and 5 represent photodetector-based and daylight sensing LED-based lighting systems, respectively [25].



Figure 4. Photodetector-Based Lighting System.



Figure 5. Daylight Sensing LED Lighting System.

Daylight harvesting systems are developed based on various methods. One of the most common methods of lighting control is by calculating the sum of illumination by daylight and artificial lighting and to evaluate the extent of fading of the light. One technique that is used is the dimming theory of PWM with partial-feedback loop control

for light fixtures that further analyzes and inputs the illuminance provided by artificial light. The main objective of this controller is to determine the illuminance level of daylight and determine the fading duty cycle founded on the idea of partial feedback and also to sustain even lighting for a series of control cycles using both daylight and artificial light. The illuminance control scheme comprises three steps namely: the measurement, estimation, and generation of the output control signal for the light fixture. For each control process, cycle, the measurement along with the calculations are done during the OFF Time and these functions are performed which consists of ON time and OFF Time, in each control output signal [26].

A more generic approach that is used to monitor lighting in any structure without any personalized programming can be accomplished by self-calibration and self-learning of the system. The resulting system controls blinds to use daylight harvesting efficiently and also resorts to artificial lighting in the absence of daylight. This system works in any location without any preprogramming involved. A WSN is used for control of lighting, which turns ON/OFF static lights and also controls the tilt angle in the windows' roller blinds. This lighting control contains three main stages-blind predictions, regulation, and control stage. For proper control of blinds, the blinds are initially calibrated for different illumination conditions. The obtained data is then used as input to the blind control mechanism. Blind prediction for the current situation is done using the calibrated data. Blind Prediction data is further forwarded to the blind control stage which adjusts the blinds in real time. Artificial light control mechanisms will illuminate the surrounding when there is no sufficient daylight or when light is not evenly distributed. Figure 6 reflects the System Control Window [27].



Figure 6. Lighting Control System.

Smart imaging sensors are used to control artificial lighting that controls numerous zones in the same room by independently regulating every one of the lighting fixtures [28]. The imaging sensor is placed on the ceiling and configured to evaluate the photometric quantities of the entire control area. Software is developed that consolidates the creation of DALI control signals for each lighting fixture, the control algorithm, and the photometric configuration of the imaging sensor. Initially, the device evaluates the illuminance levels of every work location and compares them with the respective set values of that location, calculates the ideal value and sends the suitable control signals to each luminaire. The DALI is generally used for controlling individual luminaires. The benefit of using it is that each luminaire is controlled individually by dimming it up or down, switching it ON or OFF. A program is customized to translate the control algorithm into DALI commands. An optimization algorithm is developed which determines the fading level of every one of the independent luminaires to achieve the desired levels and consistency. Daylight harvesting is employed as an innovative approach to providing energy efficiency through the development of a control system by regulating the lighting conditions for plant growth. The device regulates the fading of the LEDs based on the light luminosity data given by

the red and blue channels sensed by the light detecting sensors present on the testbed, bearing in mind the optimal red-blue mixing ratio of light essential for the growth of plants as shown in Figure 7 [29].



Figure 7. Test Bed of Lighting System.

Similarly, a multi-input and multi-output control system integrated with daylight harvesting control are used to develop an energy-efficient greenhouse lighting. The main motive of this controller is to vary the intensity of dimmable LED fixtures for achieving the desired illumination level. The proposed control device is tested in a grow-tent containing fading lamps, LED fixtures and photosensors. Fading commands given to the light fixtures are taken as control inputs and the light irradiance levels at target locations are the output variables of the MIMO control system. This system is further used in the green tent to confirm the effect of higher illuminance of daylight in plants. The results indicate that the system achieved desired DLI for specific plants and was able to maintain perfect red and blue ratios [30].

A new and upcoming method is the use of smartphones to control home lighting systems. The growth in smartphones has tremendously contributed to the development of a smart home lighting system with daylight harvesting capabilities. This device comprises a luminaire with RGB channels which is powered by an Arduino, an Android app in the users' smartphone, and a Raspberry Pi processor. The mobile application serves as a user interface to monitor the lighting device and to receive data from the light sensor in terms of lux values on the mobiles so that the closed-loop lighting input is either dimmed or brightened continuously to preserve the required amount of illumination [31]. The luminaire includes RGB.

The gateway communication between the controller and the smartphone was protected using PKI as suggested by the IEEE Standards Association and the communication between the key controller and light fixtures is protected using the AES. Moreover, to ensure that the daylight harvesting process does not interfere with the usage of the phone, an algorithm is developed using the accelerometer of the smartphone. Therefore, this system is tested and proven to give high accuracy while enabling users to perform daylight harvesting using their smartphones without compromising their desired lighting color. Figure 8 shows a block diagram of the smart lighting system [31].

For minimizing energy consumption and to find the best suitable daylight harvesting system, a typical Greek high school classroom was used in [32]. This paper evaluates a variety of illumination methods along with two devices for daylight harvesting. To estimate light adequacy and energy savings, the first system uses a stand-alone photo sensor per luminaire, while the second system uses a photo sensor per control field. This process was illustrated using the dimming curves of the LEDs calculated along with the installed photo sensor power. The results indicate that the actual 90.5 kWhp/m² of annual primary energy consumption for lighting can be reduced to 0.55 kWhp/m^2 . When it comes to the selection of the photosensor control system, comparing system 1 and system 2, the results show that

more energy can be saved while using a central photosensor. Ref. [33] ADHS is a derivative of the traditional daylight harvesting methods. In this paper, the active daylight harvesting method is extensively studied with solar concentration, collimation, and beam orientation. A 0.1 m diameter light pipe can emit an average of 30,000 lumens of daylight over 8 h each day. The ADHS system uses a high-pass mirror to filter out unwanted infrared radiation, which acts as the heating load for HVAC. Also, slope error effects, tracking error, incident angle and solar divergence angle are monitored by ray tracking in ADHS system. The biggest drawback to ADHS systems is that the effective lighting time for ADHS is more than double that of the non-tracking typical lighting system but this is compensated by an increase in transmitted light by 100 times than other systems. Other advantages of ADHS include continuity of beam patterns across solar elevations and efficiency of light extraction and control [34].



Figure 8. Architecture of Smart Home Lighting System.

The implementation of DLCs ensures major advantages such as maximizing both energy savings and visual comfort for occupants. This paper aims at conforming this by performing daylight illuminance measurements at a workplace and the ceiling in a side-lit office building during summer and winter. The model building was planned with two windows. Measurements, orientation and functioning of the system in closed-loop were modeled to observe results at both the seasons. The results showed that the DLCs have better efficiency for analyzed situations [35].

This method comprises the use of millions of micro mirrors that can guide and control the light which in turn reduces the power consumption and improves the lifetime. The concept of Optical MEMS is used and the micromirrors are electrostatically actuated and miniaturized, hence they are not visible as individual mirror elements at distance more than 20 cm. The main advantage of this method includes:

- Long lifetime despite using mechanical components.
- Lower power consumption compared to other active systems. Light is almost completely reflected inside or outside the room.
- Does not change the spectrum of sunlight.
- Works even at low temperatures [36].

A smart way to conserve energy is to combine LED lighting systems with a highresolution sensor system and a localized light sensor. The localized light sensor mounted on each fixture is capable of improving local spatial conditions and gaining an additional 35 percent savings relative to a single photo sensor monitoring all the luminaries. Moreover, an increased number of occupancy sensors are also used because of which the detection of the occupants is more reliable and hence the lifetime of LEDs is improved as they are not affected by the increased switching frequency. Savings amounting to 79% was achieved when a combination of daylight harvesting and local occupancy data was used. Moreover, the average energy savings of approximately 28% can be attributed to daylight harvesting depending on the season, window, location, window size, etc. The automated daylight harvesting was carried out by first calculating the readings of the photosensors co-located for each LED luminaire. At each phase of the system, the readings of the photosensor was analyzed and the quantity caused by electrical illumination was subtracted from it. The quantity remaining is due to daylight from which daylights contribution to the desktop level in that zone was estimated. The amount of electrical lighting needed to sustain the required illumination can be determined by subtracting this value from the target desktop illumination value [37].

Daylight lighting control systems were expected to be in higher demand given the widespread awareness of the advantages of daylight in an indoor environment and the energy-saving benefits. However, this did not happen since their functioning cannot be effectively determined while designing and that the difficulties in calculating energy and economy benefits from the daylight system. Hence a new methodology is needed to overcome these obstacles. The advantages of the new performance parameters (Daylight Integration Adequacy, Percentage Light Deficit, Percentage Light Waste, Percentage Intrinsic Light Excess) should be understood [38].

In the more recently developed buildings, a special emphasis is being given to energy savings and the quality of lighting in the building. A special case of the lighting quality and savings for the Construction, Design and Building Environment of the Beirut Arab University is taken and parameters such as Window Size, Obstruction Angle, Glazing Visible Transmittance are taken. Firstly, an analysis was done for understanding the existing lighting in the area using the Dial DIALux software, the existing situation relating to daylight was done using the Autocad Ecotect software and the information regarding the user behavior was analyzed using Hobo loggers. Based on all these data daylight designs which are based on hollow prismatic light guides were proposed which provided solutions regarding the environmental, technological, energy savings and green buildings technology needed. The proposed designs were used to obtain high intensity illumination levels and steady lighting in the facility [39].

The concept of the ASF as a scalable, effective, and versatile type of smart facades could be a potential solution for a comfortable environment through accurate integration with the parametric architecture. A study is therefore intended to analyze the production phase of ASF focused on parametric modeling techniques with an emphasis on its comfort parameters via a tractable shading framework. To optimize the advantage of daylight for individuals, the indoor architecture for daylight harvesting must ensure the visual comfort of the occupants. Daylight decreases the use of electricity, and visual convenience is usually preferred by consumers over energy quality issues for heating and cooling. Therefore, the system will adjust the shading equipment depending on the occupant's sensation of visual comfort. In this regard, the building fenestrations are responsible for incorporating daylight in indoor rooms. An origami-based hexagonal adaptive solar skin was fabricated using a single office workspace as a test bed for this study. A pragmatic strategy was systematically used in which three-time curves were developed as test methods for the study of pointin-time illumination and glare probability. The geometric characteristics of all timings were obtained from field measurements and used in simulations to increase glare change quality. The results reveal that the use of spaces with a high degree of transparency has a favorable impact on the illumination, but also a negative effect on the glare only in selected fields of indoor view. Eventually, it can be altered by choosing a proper time-based skin configuration for glare-free daylight. The findings also indicated that incorporating the proposed timing patterns over the receptive skin could enhance incoming daylight and shading opportunities [40].

The need for environmentally conscious buildings has risen over the past few years. The electric power demand of luminaries can be greatly decreased by integrating daylight harvesting, but the solar heat gain is increased. The need to analyze the solar heat gain and the natural lighting of façades using different external sun shading is required. The LCEM constructs a simulation model that tests the use of electric power by integrating Radiance's measurement results with NewHASP using the BIMs model and a heat gain calculation [41].

In [42] an embedded photometric device to combine high-resolution sky lighting tracking based on HDR imagery with quasi-real-time on-board daylight computing, consisting of a low-cost image sensor and an FPGA microprocessor is proposed. The intentional calibration of the entire imaging system with regard to its spectral response, vigñetting effect and signal response was formulated and confirmed experimentally. The instrument was designed to measure a large spectrum of luminance, including direct solar disk, sky vault, and landscape at the same time. Finally, experiments were carried out under prevailing clear and overcast sky conditions to assess its efficiency, both qualitatively and quantitatively. The built-in photometric device has demonstrated its benefits in real time daylight simulation, response time, simulation efficiency and adaptability. In the context of building automation, the device can potentially be used to control shading, illumination or electro-chrome glazing to control daylight penetration. Its ability to monitor solar luminance and location monitoring makes it suitable for producing solar photovoltaic power or changing the angle of profile of the modules [42].

Fenestrations are the reason for a substantial amount of energy consumption in the urban environment and have a significant impact on daylight and occupant's visual comfort. The equilibrium between energy usage and occupant well-being can be accomplished by automatic dynamic indoor shading systems. In this analysis, full-scale experimental evaluation of two types of roller shades using two different control systems was carried out in three separate orientations under different sky conditions. Each orientation used a typical room with no shading or lighting control, and a second similar room with embedded shading and lighting control. Two different types of glazing were also used to assess the performance of roller shades when used along with different types of glazing. Using dynamic roller shades and lighting control, relative to the baseline case of no shading system or lighting controls, an overall lighting energy saving of 52 percent was achieved. This energy conservation in lighting was remarkably robust across all test cases due to the configuration of the control strategies used, regardless of adjustments in glazing, orientation, and shading equipment [43].

Day-light-quality strategies take effective advantage of natural daylight supply and operate only a minimal amount of artificial illumination to meet the necessary level of lighting. The two primary techniques for regulating illumination in daytime areas are daylight switching and daylight dimming. Gradual lighting regulation will save 30–40 percent of electricity. The saving can be even more so with the on/off light control [44].

Daylight distribution and solar energy harvesting are two important construction strategies for mitigating energy usage in buildings. Effective daylight increases the amount of usable natural light in a room and compensates for the need for electrical lighting. This text mentions the architecture and the experimental demonstration of the LFPL method. Daylight redirection is accomplished by the geometry of the prismatic louver and accurate orientation, while energy harvesting is accomplished inside the prisms through IR absorption from the liquid (e.g., water). Daylighting efficiency was assessed at key space locations by illuminance measurements. The experiments show that in two separate systems, effective redirection can be accomplished. First, with all prismatic louver components in a fixed orientation, distributed light redirection efficiency toward the ceiling is increased 8 times. Second, by allowing various rotations of the individual prismatic louver modules, a selective dynamic redirection of the incoming solar radiation is achieved, leading to a rise of 100 to 200 times of the concentrated daylight redirection on a single ceiling area relative to the situation where no LFPL is used [45].

An LFPL that can harness daylight and heat energy with the ability to give improved degrees of natural lighting to office spaces. Daylight harvesting is one major energy con-

servation technique. For the purpose of providing natural illumination and to potentially balance electrical lighting, daylighting is the regulated entry of natural light into a building. This paper focuses on the daylight benefits of the LFPL facade approach and shows that not only can we obtain significant daylight penetration of solar radiation deeper into the office space with a reasonable range of prism louver orientation, but also lighting uniformity and potentially mitigate glare effects [46].

Parametric architecture can be used to improve construction quality by integrating and arranging design elements. In this research, daylight within the building has been improved by creating a kinetic shading system with different units that react parametrically to sunlight by 3D rotation (around the center of the units) and 2D motion (around the surface of the shading system). Various patterns have been determined to construct a basic model of system to enable the designer to make a wide range of choices. Their rotation was parametrically governed by the direction of the sun and the weather to provide constant, balanced sunlight inside the house. A simulated room and the geometry of the shading method have been built in the parametric modeling environment. Various parameters such as length, width and height of the room, the height of the upper window and the top of the window, and the dimensions of the shader units were used to calculate the effects depending on the position of the sun and the pattern of the units. Models for daylight simulation were provided on the basis of the parametric design of the shading system. A daylight simulation engine measured the UDI. In terms of the changes in UDI, the experiment was reviewed, and important improvements were found. The suggested shading scheme has shown many benefits, including the potential to maximize visibility within the room and enhance the efficiency and uniformity of daylight. The environment (diffused) light was maximized in this system and the probability of glare was minimized [47].

In this paper, light sensors and temperature sensors with solar energy harvesting and wireless connectivity capabilities have been developed. The Hybrid sensors are designed by combining printed sensing elements with EnOcean technology that delivers solar power harvesting and remote monitoring functions. The Light sensing and temperature sensing components have been printed on lightweight PET films using organic photosensitive materials and particle-filled polymer inks. The resulting hybrid sensors have been tested and shown to meet the design requirements. Light sensors have been mounted in a full-scale office test bed to show effective light management [48].

The study was conducted on the first zero energy building in Southeast Asia. It was redesigned from an existing building and comprised of passive and active construction techniques for sunny tropical weather. It was found that efficient lighting and air conditioning systems were the most successful among the active strategies. Controls were found to be most efficient among the passive technique of lighting pies and lighting. Passive tactics were considered to have a happier life and payback. It was concluded that active and passive techniques must be integrated into the architecture of buildings for increased energy efficiency. Specifically, passive architecture must be integrated as they have substantial energy enhancements [49].

The lighting of a setting has a significant impact on the working of the office workers, but the psychological variables relevant to the office setting in the process of lighting commissioning is lacking. The research examined how many workers employed in a recently renovated building felt that they could monitor the current lighting system and their perceived efficiency levels and an affective organizational contribution to explore the association between the variables and the level of satisfaction with the lighting commission process. The satisfaction with the commissioning process did not correspond with the perceived efficiency, controllability, affective organizational engagement, or the number of working hours on average after the refurbishment. Perceived efficiency strongly correlated with controllability and affective organizational commitment, both of which were associated with the perceived number of productive hours of work. The findings have also demonstrated that lighting controllability plays an important role in enhancing workers' perceived efficiency and job satisfaction [50].

More and more effort is being made to minimize building energy and increase occupant comfort daylight. The downside to this is that daylight can also cause thermal and visual discomfort. Hence, The Vertical Greening Device can be used on the outside of the building exterior and can be used to monitor the amount of light inside buildings. Plant surfaces can reflect and diffuse direct light. To verify the three typically used VGS vines were selected and their effect on the quantity and efficiency of daylight was investigated. Lighting parameters, such as illumination, refraction, discomfort, and light color temperature, were calculated and analyzed. The results showed that the glare of discomfort was reduced with the aid of the vines and also depends on the location of the sun, such as the altitude and azimuth angles, at various climatic conditions [51].

Most office lighting systems are fitted with occupancy sensors to reduce energy consumption. Since these sensors have limited reliability in the presence of detectors, very conservative strategies are typically used. In this paper, we suggest a novel lighting control method for an office that considers the performance of the sensor to be a noisy observation of the actual occupancy status. Instead of either flipping the light on or turning the light off the device dims the light in the office according to the likelihood of occupancy earned. According to this occupancy likelihood, the device optimizes the degree of lighting in the office to minimize energy consumption. Simulation results show that it saves more energy compared to the traditional "on/off" approach [52].

A sustainable lighting system in a building can easily be accomplished by feeding lighting loads to any renewable energy generation system, but the reduction of the full payback period depends in the performance of the lighting system. The goal is to decrease the lighting load to such a small amount that the payback period is limited and the building can use free of charge lighting. This research paper documents a range of initiatives that can minimize the demand for lighting in a university building and describe the implementation of these measures on the building site. In conclusion, the payback time of the planned efficient lighting system implementation is estimated to explain the need for an energy-efficient lighting system in buildings [53].

Smart room lighting is achieved using external fenestration and central and dispersed light sensors. The proposed dispersed light illumination system had an improved efficiency of 28% relative to the proposed central light illumination system. The level of illumination of the indoor atmosphere is calculated using light sensors and this electrical input from the sensors is translated to information on illumination. This light information can be calculated using a centrally placed sensor or several sensors in a dispersed fashion. This value is compared to the ideal reference value of the illumination and thus the illumination degree of the LED luminaires can be increased or decreased to obtain the desired illumination value [54].

The key purpose of the paper is to explore the effects of various design options on electric lighting energy needs and to determine the impact of energy demand for electric lighting on global energy efficiency. A simulation is conducted to explain how the supply of sunshine and the consequent energy requirement for illumination, heating and cooling differ as the design features of the building/room vary. As a case study, a single room was used and research was done by modifying its location, orientation, etc. The architecture approach for daylight optimization has proven to have a major effect on the global demand for electricity in a vacuum. Increasing the amount of daylight results in a decrease in global demand for primary energy, especially in the presence of a sensitive dimming device for daylight [16].

Daylight can be added to lighting systems in buildings to minimize energy consumption. The goal of this research is to study the effect of daylight illumination on each cycle. The illumination is measured with the daylight function of the DIALux software. The fluorescent T5 tube lamp is used for simulation. The case study room without daylight is contrasted with the case study room with the daylight of each time span. The outcome of the simulation showed that the daylight could increase the average illumination of the work plane, but uniformity is difficult to maintain. To improve uniformity, the artificial light dimming capability must be implemented to monitor the illumination of the work plane at the desired value [55].

The purpose of [56] is to determine the amount of POF required to achieve optimum indoor illumination by light transport and to explore the benefit of two-dimensional solar tacking and light concentration in indoor light enhancement. A dual-axis solar trackerbased microcontroller was developed to measure sunlight at the POF collector node every 10 s. The purpose of this paper is to decide the amount of optical fiber needed for daylight transport in order to achieve comfortable indoor lighting. The optimum amount of POF necessary for the sample space was determined by the size of the room, the location and area of the current window, the height of the ceiling and the glazing material at the window. The payback time was calculated for the study room and was judged to be within a promising range of 5.07 years. This promising technology, for further growth it is important to address the basic challenge of reducing the price of low-grade POF using dedicated supply and large-scale development because the cost of the available POF is the major part of the investment in these daylight retrofits.

3. Daylight Prediction

Daylight prediction is the process of estimating daylight metrics well advanced in times that these parameters can be used by lighting controllers to use maximum daylight illuminance and in turn reduce energy consumption. Daylight prediction data is used to control shading devices, lighting control systems which provide the user to experience an even illuminance. Daylight Harvesting is made less complex and effective by using daylight prediction techniques. Since daylight is non-linear, various prediction models are used such as time series prediction, nntool, nftool, and neural networks to achieve the most efficient and perfect prediction method to perform daylight harvesting and to obtain ideal visual comfort [57–61]. To add upon it, prominent energy forecasting models especially used for predicting solar radiation can be extended for forecasting the daylight availability. A detailed review on the available various energy forecasting models using big data and deep learning models is presented in [62].

Dynamic daylight simulations are frequently used to aid daylight prediction models. It is very difficult to predict daylight metrics at a particular time corresponding to weather conditions and occupancy levels at a target location. For this purpose, dynamic daylight metrics and dynamic daylight simulations are used in [59].

Another popular simulation method for predicting daylight is APPSM [60]. APPSM is used to predict sensor signals for controlling the lighting fixture under dynamic daylight conditions. In [60] along with the theoretical foundation of APPSM, validation of results from APPSM was also performed to validate the accuracy of the prediction results. Prediction results from APPSM were validated using PDENS program results. APPSM algorithm works by the principle of daylight coefficient principle and hence it assumes the illumination from sun and illumination from the sky as separate light sources. APPSM algorithm uses daylight coefficient values of 145 sky patches for evaluating the illuminance contributions from patches to the target including interreflections. The 145 sky patches structure is shown in Figure 9 [60]. The association between the luminance of the sky patch and the intensity at the target point is called the daylight coefficient. To gain better precision when calculating daylight, daylight illumination is subdivided into direct component and reflected component. Rtrace-a radiance tool is used to compute direct illumination from the sun. The reflected illuminance component is calculated using the daylight coefficient data that has been predicted using 145 sky patches which were used to calculate illuminance contribution from the sky [60]. Radiance and Energy Plus software is the most efficient and reliable software used in daylight simulations to predict daylighting and energy usage. A simulation-based lighting model is proposed in [61], which uses prediction methods to speedup computation.



Figure 9. 145 Sky Patches Structure.

The three-phase method from Radiance software was used for daylighting simulations since it employs BSDF algorithms to avoid complexity and hence it speeds up simulation time. These predicted values are then fed to respective control systems for the desired use of daylight [61].

Three machine learning algorithms are evaluated in [63], which are PCA, ANN and SVM. The Overall structure of prediction followed by the algorithms is shown in Figure 10. Input needed by machine learning algorithms such as weather data, building design data, and window configurations is given by the data generation programs such as Python and MATLAB. The algorithm then predicts the hourly daylight levels at several sensor points and also estimates the hourly energy consumption. UDI is used as a daylighting metric in [63]. The algorithm classifies the daylighting level values into different UDI values and also estimates the energy consumption. When prediction results from PCA, ANN, and SVM are compared with real-time data, it can be observed that the prediction data generated by PCA has the highest accuracy when compared with others. An algorithm cannot be run using all of the parameters because it will slow down the algorithm's performance and make it difficult to visualize so many features in any graph. As a result, the number of parameters provided to the algorithm as input must be limited. It is difficult, frustrating, and time-consuming to manually find correlation in thousands of parameters. This is something PCA excels at. All of the Principal Components are independent of one another after applying the PCA. There is no link among them. With limited parameters, the training time of the algorithms is greatly reduced. Values predicted by PCA can be further used by lighting control, blind controls to obtain the desired illumination [63].

Similarly, Ref. [64] employs ANN for daylight prediction. In prediction models, to estimate interior illumination due to daylight, the illuminance of daylight incident on the exterior of the window must be estimated. Ref. [65] proposes an efficient daylight prediction model that combines the Perez Model to estimate exterior illuminance and uses the ANN model to predict interior illumination. This prediction model with ANN provides fast, accurate, and reliable values of illuminance.

The Kriging method is frequently used for various prediction and interpolation problems [66] uses the kriging method coupled with energy simulation tools to predict daylight conditions. This system when combined with an energy simulator can plot hourly indoor light illuminance.

BSDFs are used in modeling and predicting daylight scattering by considering building window parameters, but BSDFs are very expensive and are not often reliable. Hence ref. [66] proposes an analytical BSDF model that predicts daylight design metrics based on climatic change. This system is efficient and accurate which is based on the empirical BRDF. The devised BSDF model is inexpensive and can model common fenestration devices such as shades and blinds. They are very effective while predicting daylight metrics at the design stage. This analytical model can be easily integrated into ray-tracing programs such as Radiance which can further be used to predict daylight due to direct radiation [66].



Figure 10. Prediction Algorithm.

The problem and the only practical way to gain deep insight into complex shifts of the cloud is by computational simulation [67]. Hence ref. [67] developed a model using the UniSky simulator and HOLIGILM tool, which provides accurate predictions of fundamental variations of light guide behavior under homogeneous and non-homogeneous cloud conditions. Natural light can be delivered into a light guide system. One such light guide system is a light pipe since it does not require any additional energy sources and does not have a negative effect on the natural environment. The structure of a light pipe is shown in Figure 11. Though several daylight prediction methods are invented, it is very challenging to accurately predict the performance of lighting control under dynamic meteorological conditions since existing methods do not account for random configurations of clouds and its high changing frequency [67] treats the light field below the tubular light pipe theoretically and numerically by taking nonhomogeneous sky patterns into account. This numerical assumption is used as input in UniSky and HOLIGILM to predict daylight conditions in the building which has light guide systems [67].



Figure 11. Structure of Light Pipe.

Light transported through internal reflections via the guidance mechanism is deliberately or accidentally spread in both directions, depending on the optical properties of the output unit. Ideally, increasing the flux density if the luminous energy that escapes the base of the light tube results in the best possible indoor illumination [68].

The lack of data to represent the energy consumption of buildings in underdeveloped countries makes it hard to decide on the intervention methodology for the existing building stock. This difficulty was tried to overcome using a 6-step process to examine and identify a methodology for the energy-saving ability of the existing building architecture. The six steps include a process to define a reference building, dynamic archetype model simulations, a baseline estimation of the energy consumption, selection of measures for the energy retrofit, an analysis of the optimal cost, development of scenarios for energy saving scenario projection demonstrated the urgency of dealing with the high-rise offices at present and that policymakers should be more observant of new buildings to reduce energy consumption [69].

An HDR image processing algorithm is employed in [70] for capturing and predicting real-time sky conditions for controlling input signals to a building's AF system, DH, and HVAC system.

The performance of three non-linear prediction models as performed in [59] are examined in [71]. NLARX, TDNN, and ANFIS are the three non-linear models that are used in daylight prediction that are evaluated in [71]. These three models use online interior and exterior measurements to predict daylight illumination. They are evaluated using the RLS adaptive algorithm, which identifies the error between the actual system parameters and results from the predictive algorithm. Initial values of interior and exterior illumination are fed into the system as input and then the models predict the illumination values *p* steps ahead. The exterior and interior illuminance are expressed as *Eext* and *Eint* respectively. (3) and (4) are used to predict illuminance values '*p*' steps ahead [71].

$$E(k+p) = f_0(E(k)) \tag{3}$$

$$E(k+p) = f_0[E(k), E(k-\tau), E(k-\tau), E(k-2\tau), \dots, E(k-(d-1)\tau]$$
(4)

The results from predictive algorithms are then verified using RLS to check which one has the least error [71].

Most of the daylight prediction models use photometric sensors to predict daylight metrics, but ref. [72] introduces a daylight prediction system through theoretical discussion followed by empirical analysis, which is a sensor-less model that employs mathematical equations to predict daylight illumination. This is done by using an exponential distribution that uses lux reading from a weather station to predict the daylight at a given target location in a building. This model allows us to estimate the dimming level required to achieve the desired illumination that uses both daylight and artificial light. For a sensor-less daylighting system, a smart lighting control system, lighting simulation software, numerical evaluation software, and a rooftop weather station to predict daylight conditions are required. To effectively maintain the lighting levels the lighting parameters are divided into external and internal variables. External variables are the daylight metric that will be predicted by the weather station and the internal variables concern the target environment design and hence they are considered to be constant. The ratio of internal lux intensity to external lux intensity is known as the DF [72].

$$DF = (Internal Illuminance/external illuminance) \times 100\%$$
(5)

The variables affecting daylight penetration into the building are unattended which brings complexity while using DF. Hence a relationship between external illuminance and internal illuminance at a particular target point is used as a scalar factor which depends on decaying luminance intensity known as SF is used and this varies with varying distance from the window [72].

$$SF = (External LUX Value/Internal LUX Value)$$
 (6)

Presently, hot desert countries are adopting policies that ignore their climate change dependent on fully glazed constructions. This results in low productivity and identity crises in construction. Any effort, therefore, seeks to minimize excess sun exposure and make available effective sunlight. The results from the observations and studies suggest that Passive bio-climatic vernacular architecture approaches and daylight strategies by incorporating smart materials can achieve clean energy, thermal/visual comfort, and daylight design objectives [73].

Daylight simulation for a real space using TMY data and active local weather data with a one-minute interval was performed in [74]. In this research TMY data was used to foresee the quality of daylight, develop a daylight-responsive closed-loop lighting control system, to select calibration hours, and to predict the efficiency of the control system. Whereas, the active local weather data was used to determine daylight quality. The purpose of this analysis was to investigate whether the traditional weather data (i.e., the TMY data) allowed reliable daylight quality and daylight-responsive performance control system predictions. A real space model equipped with four different types of fenestration systems such as clear glazing, 0 degree and + and -45 degree venetian blinds and three different daylight simulation methods such as daylight coefficient, three-phase and five-phase methods were used for comparison in this research. It was observed that the expected energy saving based on the TMY data was accurate when the calibration was carried out in environmental conditions close to those defined based on the TMY data. Conversely, calibration under inappropriate environmental conditions has been shown to result in lower energy savings.

4. Influence of Building Geometric in Daylight Harvesting

In recent years, the advantages of using daylight and the plans of many buildings taking daylight into account have increased exponentially. The architectural influence makes daylight a vital aspect of the construction process. The incorporation of daylight in architecture, by using design methods, results in better daylight and thus reduces visual discomfort. The key goal is to replace electric light fixtures with natural daylight and to reduce the expense of electric charge. Hence apart from lighting control modules, building design can also be used to achieve the desired illumination using daylight [75,76].

The architecture and orientation of model buildings for the different climatic zones of India is analyzed in [75]. The target building is modeled using the Ecotect daylighting tool and simulated with the help of Radiance Beta V2.0. [75] focuses on the study of interior lighting due to daylight availability in traditional low-rise buildings under various climatic conditions. Daylight availability analysis aids in perfect designing and will also help to minimize artificial lighting sources.

As the model building is considered to be in India, climate variables such as solar radiation, ambient air temperature, air humidity, wind speed, forecast, etc. have been taken from the Bureau of Indian Standards. To evaluate different climatic conditions, the model building is simulated to be in different cities with varying climatic zones.

Equinox is a phenomenon in which the earth's equator crosses through the center of the sky, making day and night more or less the same with respect to length. Equinox takes place every year in India on 21st March and 21st September, daylight analysis on equinox days is performed for the cities considered. The area with daylight illumination and the area illuminated artificially differ from dusk to dawn, so equinox is used because it has equal periods of day and night so that we can achieve the average value of the illumination effects. The area covered by the selected level of luminous intensity is separated from the simulation and therefore the average area of illumination is estimated. Results from [75] show that the area illuminated by daylight of the model building is maximum when the building is located in a temperate and least in a composite climate environment [75]. VELUX is also like the Radiance Building Simulator used to design building models so that they can be validated for obtaining maximum daylight illumination. This is used to test construction models for estimating daylight metrics such as daylight impact and

light distribution. The daylight factor is known as the ratio of indoor lighting and outdoor lighting and is typically written in percentage form [76].

In [76] daylight impact and light distribution relating to various space conditions and window settings were examined. Daylight can be used well because it falls on the windows every day. Several criteria are weighed such as construction, the height of the door, sky views, reflectance from the ground, and interior design, to achieve viability for each portion of a room. Usually, credits for daylighting are given in various iterations of the LEED (Leadership in Energy and Environmental Design) rating systems and are given in correspondence to room architecture. This credit system determines the criteria and their application to satisfy basic specifications. A building architecture strategy should improve the ability to achieve the required credit criteria. For example, windows placed at or over 30 inches above the floor level should satisfy the following requirements [76]:

$$0.150 < V LT * WFR < 0.180$$
 (7)

From Simulation performed using VELUX in [76], it is obvious that the configuration of the window influences the distribution of daylight in the house. The daylight factor increases when window size increases. Simulations were also performed for analyzing the perfect window type among punched, strip, and full windows. The tests performed show that the penetration range of daylight is greater in the case of strip windows relative to the punched ones. It is also noted that strip windows achieve uniform daylight in the room [76].

A parametric study of the impact of independent variables on dependent variables with the help of computer simulations using RELUX software is performed in [77]. This scientific study analyzes the impact of various architectural elements such as reflectivity of surface materials, the geometry of space, internal layouts, dimension, and position of the windows as independent variables on the dependent variables such as uniformity rate, daylight factor. Several configurations were simulated using Relux software to understand the role and influence of architectural elements in a study space for a better and more balanced distribution of natural light. The results have been summarized as follows:

- Windows with a maximum height to width ratio of 35 to 45% with the preferable ratio of 40% can decrease the electrical energy consumption by 32% with horizontal and vertical divisions. It also improves visual comfort by 7.5%.
- The optimal length to width and height ratios are 6.5 and 5.2 respectively, which reduces the electrical consumption by 34%.
- In terms of inner surface layout, using horizontal and vertical surfaces with two front windows along the length of the building provides a better uniformity rate in lighting [77].

A similar study was performed in [78] using DIVA software to research the impacts of various rooms and window geometrics on daylight harvesting. Two classrooms were simulated, a square and a rectangular classroom of the same total space and they have a square and a rectangular window with 8% and 12% WFR, respectively. The rooms have two separate dimmable lighting schemes, fluorescent and LED dimmable lights, to measure energy savings. The measurement of daylight is carried out by the assessment of Daylight factor (DF) and Daylight autonomy (DA). The results from the DIVA simulator have been summarized as follows:

- At constant WFR, a rectangular room achieves better energy performance with a rectangular window. Similarly, for a square room, the best performance are provided when equipped with a square window.
- DA ratios do not show any noticeable differences once the orientation and WFRs are fixed.
- The best energy result is obtained when a combination of a rectangular room with a rectangular south-oriented window, with 12% WFR [78].

To find an easy and objective way of daylight harvesting, performed a daylight simulation to acquire basic data about factors affecting daylight harvesting. A simple simulation model has been developed and was assumed to be in Tokyo with the following dimensions: 39.8 m wide, 15 m depth, and 2.8 m high, and it was assumed that the modeled layout office was fitted with daylight harvesting systems using ceiling sensors. Different combinations of Window-to-wall ratio, window orientation, no of windows, Venetian blind position, and depth of space were used to study for daylight data, which combined annual solar light and skylight distribution per 1 h in the Tokyo metropolitan area. It is evident from the simulation findings that there is a strong link between window-to-floor ratio and annual energy consumption of artificial lighting and it is independent of space form and window orientation.

Approximately 10% of electricity can be reduced when the window-to-floor ratio is 10% and consumption can be reduced by 20% when the aperture ratio is 20%. It was also observed that the conservation benefits of daylight harvesting improved by a factor of 2 when automated monitoring of the Venetian blind system was used. Hence it is clear that the Window-to-Floor ratio is one of the easy and objective ways to assess the energy-saving effect of daylight harvesting [79].

The Atrium is an important and efficient architecture used for using maximum daylight. Daylight use using an Atrium is one among the easiest ways to boost energy quality and indoor atmosphere.

The design of an atrium can be defined using a parameter known as the plant aspect ratio (PAR). PAR is the ratio of the width and the length of the atrium [80], which determines the elongation of the atrium as expressed in (8).

$$PAR = (Width/Length)$$
 (8)

To figure out the correct configuration of the atrium's PAR, various models of atrium width and length were studied in [80]. The models were simulated using Radiance and values of Average Daylight Factor (ADF) were also estimated for each model. The study indicates that the expansion of the atrium's length would allow for more sunshine in offline spaces with the optimum mode of PAR as 1/3. The findings further show that the increased height of clerestory windows will increase the amount of ADF in the atrium and adjacent spaces [80].

Numerous studies established that students' and a teacher's capability to teach, learn and their general health are heavily dependent on the standard and quantity of daylight and indoor thermal conditions. The main purpose of using daylight in schools is to minimize energy usage and costs, but also to enhance student efficiency. An appropriate window configuration increases comfort by reducing glare, spreading light, and regulating the solar energy gain. This investigation's focal point is on the effect of various transparency ratios and window designs in two vital directions (west and east) on the comfort of the occupants and the energy requirements of the classroom. A structure was chosen for an investigation at the design stage. One classroom facing the east direction and another facing the west direction have been studied. The two classrooms were simulated using the lighting and energy simulation programs DIALuxEvo 6.0, DesignBuilder 5.5 and EnergyPlus 8.9. The study aimed on the stage of architectural design and is therefore dependent completely on simulating a model virtually and other required calculations. The main goal was creating a design in which daylight would complement artificial light when it was not enough. A total of three different models, based on different transparency ratios, are developed to describe the effect on the indoor conditions of classrooms. Based on the study of comparable buildings, the window height was fixed at 1.70 m and the sill height was set at 0.90 m in both versions. A 50% glazing ratio was found to minimize the need for artificial lighting by at least 15 per cent and provide more comfortable conditions [81].

The effect of the extended metal mesh on the building of energy consumption and natural daylight is also discussed. Parameters including WWR, perforation rate, and

window glass glazing were examined and the daylight standard of the LEED rating system was used for evaluating. In this analysis, three variables were widely used in building facade design to address the effect of lighting and air conditioning on energy consumption. From the lighting simulations, it can be concluded that only three scenarios were compatible with the LEED daylight standard and S16, consisting of 50 per cent WWR, laminated transparent glass and extended metal mesh at 21% perforation, showed the highest energy efficiency. It has been found that WWR plays a key role in affecting the quantity of daylight entering the house. As a result, higher the WWR, more is the energy-saving capacity of extended metal mesh and glazing [82].

The adjustable window awning shelf contains a canopy connected on both sides of the window to a support. The supports are attached to vertical drive screws allowing for the capability of pushing the canopy up and down. Each drive screw is connected to a standard driveshaft. During the time of the year during which cooling is required, the canopy is spread out above the window, shielding the window from the light. During the time of the year when heating is required, the canopy is placed at the bottom of the windows by rotating the supports with the canopy down. The canopy angle relatively increases with the window angles. When in the lower position, the awning reflects the sunlight falling on the window and increases the quantity of sunlight and solar heat entering the building through the window [83].

This research investigated the nature and daylight efficiency of a new Louver screen for office buildings. A Louver device capable of providing sunshine and thermal control efficiency as a part of an environmental façade strategy for office buildings was aimed to designed. The Louver screen was assessed for three distinct material finishes such as conventional materials commonly used in louvers and two other types of materials with ceramic finishes with an effort to decrease the system's environmental impact. Annual efficiency simulations for the full-scale room, using Daysim software lighting tools, were performed to gauge the impact of the three material systems on indoor daylight levels and distributions using both the standard daylight factor and Daysim climate-based daylight metrics. The key purpose was to examine the possible lighting efficiency of the louver device by investigating the various geometrics in the louver profile. This paper states that daylighting systems can be classified based on three main features:

- Daylighting function—light distribution performance, visual comfort, and solar gains control.
- Building integration—interior, exterior, envelope of the building.
- Type of operation—passive or active.

Each type has various features that need to be taken into account while constructing the louver scheme. According to observation, a louver with a specular surface has been shown to be a daylighting device capable of achieving optimum daylight efficiency, particularly under a radiant sky with sunlight. The results indicate that louver with traditional material can provide an acceptable degree of daylight and visual needs inside the room, whereas ceramics tend to be a viable substitute medium to be used in the development of advanced daylight technologies [84]. The need to use natural and green energy sources to power buildings has brought growing attention to daylighting issues over the last few decades. The daylighting architecture in ancient Romania was examined for this purpose. The work was split into two sections: a typological study of Roman houses, aimed at explaining the relation between the parameters for ancient architecture and daylight harvesting and a dynamic simulation of the daylight conditions. Ancient architects were very concerned with daylighting, experimenting with design methods, mostly very basic, but successful in providing useful light conditions for everyday indoor activities. The appearance of the building form and facades was strictly related to both the outside environment and the activities carried out inside. This correlation between the building environment and the atmosphere helped to determine a particular culture of architecture and therefore a prudent use of daylight. To explain simulation results such as the CDI, the corresponding minimum CDImin, and sCDI, a new set of performance metrics was created. The CDI at

the point is the maximum task illuminance of the collection for which a daylight autonomy equal to at least 50 percent is measured, provided a point and set of task illuminance values (e.g., those suggested by the standards). Then, given a CDI, the proportion of the floor area characterized by this very CDI value is the corresponding sCDI. For example, in a room where sCDI 300 lx is equal to 50 percent and sCDI 500 lx is equal to 50 percent for 50 percent of the year, the daylight guarantees an illuminance value equal to 500 lx in half of the room alone, while a 300 lx illuminance is guaranteed in the remaining part of the room. Lastly, the minimum CDI is the minimum CDI value observed in the area, i.e., the value for which an autonomy of at least 50 per cent of daylight is observed in the entire room. The results seem to indicate that there was a strong relationship between the levels of indoor sunlight and the operation of the rooms. The more sunlight a room earned, the more impressive its function was [85].

Lighting strategies can be applied to create environments which are friendlier to the occupants and reduce the interventions in lighting conditions. An IVR environment can be used to test the design system quick and easy. The rated contrast on the window wall as well as the negative lighting interventions in an IVR office room with different window-to-exterior-wall ratios can be increased by increasing the illuminance of the areas around the window and the electric wall-washing system. In relation to other lighting conditions, the greater lighting contrast between the window and the surroundings was found to be 15 percent WEWR. The suggested low-power electric wall-washing device will decrease the possibility of the consumer interfering in a room with different size windows with the lighting conditions [86].

The human biological clock is mainly synchronized by the perceived light, specifically the short-wavelength daylight rays, also known as the circadian rhythm. Human health and well-being may be threatened by inadequate access to daylight or comparable electrical lighting. The results of circadian stimulus autonomy, i.e., the percentage of days during the year when CS is above the minimum threshold in conventional classroom designs, are seen in this study. It is quantified by circadian stimuli, facilitated either by natural or electrical illumination. Under three standard sky conditions, the location studied has a window of variable length, direction, and orientation, as well as different reflectance values of the classroom's inner surfaces. The goal of this study is a first approach to the determination of the appropriate window size for multi-purpose educational building classrooms, in order to promote the correct CS value for students and to provide a well-trained circadian system, as well as to evaluate the effect of electric lighting on their biological clock. Taking into account the different daylight SPDs and the variable luminance of the sky vault, measurements were made for three positions. The classroom studied had a variable window size with two primary orientations and a variable reflectance of the interior surfaces. The following results were obtained:

For different conditions and variables, such as the frame size, the reflectance of the inside surfaces and the observed working plane, as well as the color temperature of the ceiling luminaires, the CS may be quantified. In the case of a light blue work plane, cool LED lamps with a CCT of 6500 K allowed up to 5 percent more CS than warm luminaires, 19 percent more compared with the results for a white work desk and 81 percent more than the light wood environment scenario. Window orientation also affects this metric's measurements, but its impact and lintel height are not as important when the reflectance of the classroom's inner surfaces and the observed atmosphere is sufficiently high [87].

5. Light Pipe Efficiency and Life Cycle Cost for Daylight Harvesting Systems

The light pipe is studied elaborately to understand its efficiency and internal illuminance distribution. For the same objectives, extensive experiments are performed for a continuous eight months considering three cloud conditions, namely clear, cloudy, and intermediate [88]. It was found that the daylight penetration factor remained almost constant during cloudy days and varies significantly during clear days. A study suggested to avoid excessive aspect ratio and many bends to obtain effective daylighting devices with light pipes. Also, larger diameter light pipe should be utilized whenever possible [89]. The prismatic diffusers are compared extensively with the radial diffusers for internal illuminance distribution parameter. It is observed that prismatic diffuser absorbs light with uniform light distribution just below the pipe, whereas the radial diffuser provides higher light under the pipe and absorbs light away from pipe for working plane that are placed 700 mm below the pipe [90]. The performance of top lighting and side lighting light pipes are studied in and it was observed that the side lightning light pipes had better illumination characteristics in morning while top lightning light pipes performs better in afternoon under sunny conditions of Beijing [91].

The cost of implementation of daylight harvesting scheme is another important aspect of daylight harvesting. Two different lighting schemes were compared wherein one scheme involves Fluorescent luminaires without daylight harvesting and other one involves LED fixtures with daylight harvesting combined with occupancy and daylight dimming sensors. A simple replacement of fluorescent lamps with LED light fixtures itself led to savings of about 22% in the Life Cycle Energy Cost. Moreover, LED fixtures have added advantages like higher lamp life and the ability to add features like occupancy control. Furthermore, when the LED fixture was combined with daylight harvesting and occupancy and daylight dimming sensors, the Life Cycle Energy Cost is reduced by 33% of what can be achieved by using only fluorescent luminaires [92]. In another study, the life cycle cost analysis for manual, interior and exterior automated lighting system was performed. It was found that the manual blinds cost the least and the exterior automated system was the most expensive one. This was probably due to the required equipment and the complexity of installation controls for the exterior system [93]. Hence, daylight harvesting systems have proven to be cheaper and more environmentally friendly in the long run.

6. Case Study of Daylight Harvesting in Urban India

To simulate the actual daylight contribution for any building before construction, ECO TECH software is used by providing the inputs such as building orientation, reflectance's values of the ceiling, wall, floor to analyze the overall percentage of daylight penetration available verses the percentage prescribed in the IGBC to obtain the credit points. To simulate the actual daylight contribution for any building before construction, ECO TECH software is used by providing the inputs such as building orientation, reflectance's values of the ceiling, wall, floor to analyze the overall percentage of daylight penetration available verses the percentage prescribed in the IGBC to obtain the credit points.

Hence if the overall percentage of daylight penetration is between 50 to 75% then the building is eligible for one credit point (total of one point), further, if the percentage is between 75 to 95% then it is eligible for additional one credit point (total of two points) and in some case if the percentage is more than 95% then it is eligible for an additional two credit points (total of three points). Refer below IGBC extracts related to daylight harvesting credit points.

In this paper, the Daylight Analysis is employed on a proposed multi-story office building in urban India. The specific objective of this study is to evaluate the daylight quality as per the requirement of IGBC Green NB: SA Credit 03—"Passive Architecture" & IEQ Credit 02—"Daylighting".

Requirement as per IGBC Green New Buildings (Version 3.0):

- SA Credit 03—Passive Architecture.
- Option 2: Prescriptive Approach-Daylighting: 50% of the regularly occupied spaces with daylight illuminance levels for a minimum of 110 Lux (and a maximum of 2200 Lux) in a clear sky condition on 21st September at noon, at working plane (through simulation or measurement approach).
- IEQ Credit 02—Daylighting.
- Option 1: Simulation Approach-75% of the regularly occupied spaces with daylight illuminance levels for a minimum of 110 Lux (and a maximum of 2200 Lux) in a clear sky condition on 21st September at noon, at working plane. With the given

information from architectural drawing and considering actual materials and clear sky conditions of Chennai, the modeling and simulations are carried out in the Design Builder software tool. Daylight analysis was performed for the IGBC recommended living/regularly occupied spaces.

With the given information from architectural drawing and considering actual materials and clear sky conditions of Chennai, the modeling and simulations are carried out in the Design Builder software tool. Daylight analysis was performed for the IGBC recommended living/regularly occupied spaces.

The following specifications are considered for the proposed building:

- Wall: Paints with surface reflectance of 50%.
- Floor: Floor tiles with surface reflectance of 20%.
- Ceiling: Paints with surface reflectance of 70%.
- Glazing specification are considered in Table 1.
- Figure 12 shows considered working plane height, sky conditions, and solar position for daylight simulation as per IGBC. Daylight is studied for the proposed blocks. Table 2 shows the daylight simulation result for Office Building.
- Figure 13 show considered minimum and maximum threshold limit of illuminance as per IGBC requirements.
- Figure 14 shows the LUX color indicators.

Calculation Options			
		~1	
Calculation Description		Ý	
Calculation Options		×	
Report type	1-Map	•	
Detail template	6-High quality	•	
Working plane height (m)	0.7500		
Margin (m)	0.500		
Sky		×	
Sky method	1-Standard sky	•	
Sky model	2-CIE clear day	•	
Solar position		×	
Month	Sep	•	
Day	21	•	
Hour	12	•	
Grid		×	
Min Grid Size (m)	0.500		
Max Grid Size (m)	1.000		

Figure 12. Daylight Simulation Input Details.

Display Options General		
\checkmark		
Report type		×
Туре	2-Grid (Radiance)	•
Margin		**
Threshold		×
Threshold type	2-Illuminance	•
Min illuminance (lux)	110.000	
Max illuminance (lux)	2200.000	

Figure 13. Illuminance Threshold Limit.



Figure 14. LUX Color Indicators.

 Table 1. Glass Specifications of Business Center.

Space	Product	VLT (%)	SHGC	U Value (W/m ² K)
Office Building	6 mm SKN 765 +12 mm air gap + 8 mm clear	38	0.24	1.5

Table 2. Daylight simulation results for Office building 1.

Floor Level	Regularly Occupied Area	Floor Area (m ²)	Floor Area above Threshold (%)	Area with Illuminance between 110 to 2200 lux (m ²)
	Lobby1	56.2	97.9	55.1
	Corridor1	100.5	29.0	29.2
	Entrance lobby1	169.3	100.0	169.3
Ground Floor	Corridor2	102.0	41.7	42.5
	Entrance lobby2	163.2	100.0	163.2
	Office space	1536.3	100.0	1536.3
	Office space	3023.5	53.2	1608.8
1st Floor	Lobby1	55.9	24.2	13.6
	Lobby2	55.8	50.0	27.9
	Office space	2993.2	64.4	1927.0
	Breakout area	83.4	100.0	83.4
0 15	Corridor2	33.8	38.5	13.0
2nd Floor	Lobby2	55.8	100.0	55.8
	Lobby1	55.8	45.8	25.6
	Corridor1	31.6	25.0	7.9
	Office space	2993.2	67.5	2020.1
	Breakout area	83.4	100.0	83.4
2 J. El	Corridor2	33.8	38.5	13.0
Srd Floor	Lobby2	55.8	100.0	55.8
	Lobby1	55.8	47.9	26.7
	Corridor1	31.6	25.0	7.9
4th Floor	Office space	2993.2	72.1	2158.1
	Breakout area	83.4	100.0	83.4
	Corridor2	33.8	38.5	13.0
	Lobby2	55.8	100.0	55.8
	Lobby1	55.8	54.2	30.2
	Corridor1	31.6	29.2	9.2

Floor Level	Regularly Occupied Area	Floor Area (m ²)	Floor Area above Threshold (%)	Area with Illuminance between 110 to 2200 lux (m ²)
	Office space	2993.2	76.4	2287.4
	Breakout area	83.4	100.0	83.4
5th Floor	Corridor2	33.8	42.3	14.3
	Lobby2	55.8	100.0	55.8
	Lobby1	55.8	58.3	32.5
	Corridor1	31.6	29.2	9.2
	Office space	2993.2	31.0	927.3
	Breakout area	83.4	100.0	83.4
	Corridor2	33.8	38.5	13.0
6th Floor	Lobby2	55.8	100.0	55.8
	Lobby1	55.8	58.3	32.5
	Corridor1	31.6	33.3	10.5
	Office space	2993.2	79.7	2386.8
	Breakout area	83.4	100.0	83.4
	Corridor2	33.8	46.2	15.6
7th Floor	Lobby2	55.8	100.0	55.8
	Lobby1	55.8	62.5	34.9
	Corridor1	31.6	33.3	10.5
	Office space	2993.2	81.0	2424.2
	Breakout area	83.4	100.0	83.4
	Corridor2	33.8	50.0	16.9
8th Floor	Lobbv2	55.8	100.0	55.8
	Lobby1	55.8	66.7	37.2
	Corridor1	31.6	33.3	10.5
	Office space	2993.2	81.9	2451.2
	Breakout area	83.4	100.0	83.4
	Corridor2	33.8	46.2	15.6
9th Floor	Lobby2	55.8	100.0	55.8
	Lobby1	55.8	75.0	41.8
	Corridor1	31.6	33.3	10.5
	Office space	2993.2	32.3	966.8
	Breakout area	83.4	100.0	83.4
	Corridor2	33.8	38.5	13.0
10th Floor	Lobby2	55.8	100.0	55.8
	Lobby1	55.8	68.8	38.3
	Corridor1	31.6	33.3	10.5
	Office space	2993.2	84.4	2526.9
	Breakout area	83.4	100.0	83.4
	Corridor?	33.8	50.0	16.9
11th Floor	Lobby?	55.8	29.2	16.3
	Lobby1	55.8	79.2	44.2
	Corridor1	31.6	33.3	10.5
	Office space	2993.2	89.0	2664.0
	Breakout area	83.4	100.0	83.4
12th Floor	Corridor?	33.8	65.4	22.1
	Lobby?	55.8	33.3	18.6
	Lobby1	55.8	100.0	55.8
	Corridor1	31.6	41 7	13.2
	Conndon	51.0	41.7	13.2
Total Regularly O	ccupied Area (m ²)	41,051.1	Total Area with lux between 110 to 2200 (m ²)	28,515.9
% of Regularly occupied daylight illun	space achieved required ninance levels	69%		

Table 2. Cont.

Based on this, Figures 15–20 shows the daylight simulation results of proposed Office Building. In our case study, the building is simulated using ECO TECH software and from the analysis the overall percentage of day light penetration is around 69% which is between 50 and 75%. Hence this building is eligible for 1 credit point stipulated under IGBC.



Figure 15. Overall Achieved Day Light Levels (110 lux to 2200 lux) on (a) Ground Floor (b) 1st Floor.



Figure 16. Overall Achieved Day Light Levels (110 lux to 2200 lux) on (a) 2nd Floor (b) 3rd Floor.



Figure 17. Overall Achieved Day Light Levels (110 lux to 2200 lux) on (a) 4th Floor (b) 5th Floor.



Figure 18. Overall Achieved Day Light Levels (110 lux to 2200 lux) on (a) 6th Floor (b) 7th Floor.



Figure 19. Overall Achieved Day Light Levels (110 lux to 2200 lux) on (a) 8th Floor (b) 9th Floor.



Figure 20. Overall Achieved Day Light Levels (110 lux to 2200 lux) on (**a**) 10th Floor (**b**) 11th Floor (**c**) 12th Floor.

7. Conclusions and Future Scope

This study presents a detailed review of all the recent technological advancements in the field of daylight harvesting. The concepts of daylight harvesting and its significance is well-established. Daylight harvesting systems and algorithms provide the next-generation solution to achieve clean energy and also facilitate tremendously in minimizing electrical consumption globally. Initially, algorithms employed for daylight harvesting such as Genetic Algorithm, Neural Network, Fuzzy logic, and PWM control for the dimming of LEDs were discussed. It was seen that user comfort and needs are also given prime importance during the process of designing these systems. It was also found that with the emergence of IoT and due to the possibility of the interconnection of daily life devices, smartphones can also be used to control the home lighting system. Various tools and methods that are recently discovered are discussed and the importance of daylight prediction is portrayed. Also, the various aspects of architecture that affect daylight harvesting are highlighted. It is found that architectural design makes a huge impact on the process of daylight harvesting. For instance, it was found that high glazing provided good lighting which could further provide good daylight harvesting. It is evident that the geometry of the window affects the daylight distribution and uniform daylighting can be achieved by using strip windows. It further talks about the impact of various architectural aspects such as WFRS, DF, PAR, and ADF, which are analyzed, and their impact on daylight harvesting was discussed. Furthermore, the daylight saving was employed on a multistoried building in urban Chennai, India and the results are very satisfactory. The results obtained comply with required daylight factor via simulation for the proposed buildings. Also, it is determined that the building achieves the IGBC Green NB recommended daylight level in more than 50% (i.e., 69%) of living spaces; hence the office building meets the criteria of sustainable architecture and design criteria Passive architecture credit: Daylighting. The building is eligible for 1 credit point stipulated under IGBC and also meets the sustainable architecture and design requirement. Using the proposed daylight harvesting scheme it is possible to realize self-sustainable buildings which is incorporated with suitable daylight energy storage features. This in turn aids in realizing sustainable smart cities for the benefit of the future generations.

Author Contributions: The authors G.S.O.V. and K.K. contributed toward conceptualization, methodology, formal analysis and investigation; K.K. was responsible for supervision of the work. U.S., R.M.E., S.H.K. and R.P. contributed toward visualization and validation of research investigation; M.R. and R.M.G. contributed to writing the original draft. G.S.O.V., U.S., R.M.E., S.H.K. and R.P. contributed toward writing review and editing. All authors have read and agreed to the published version of the manuscript.

Funding: The Article Processing Charges (APC) of this publication is supported by the Prince Sultan University.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments : The authors would like to thank the School of Electrical Engineering, VIT Chennai, India & Larsen and Toubro Pvt. Ltd., Chennai, India for the technical expertise provided. They also acknowledge Department of Communications and Networks, Renewable Energy Laboratory, College of Engineering, Prince Sultan University, Riyadh, Saudi Arabia; Department of Electrical and Electronics Engineering; Clean and Resilient Energy Systems Laboratory, Texas A&M University, Galveston, TX, USA; Department of Engineering Management, College of Engineering, Prince Sultan University, Riyadh, Saudi Arabia for the technical inputs provided. The authors would like to acknowledge the support of Prince Sultan University for paying the Article Processing Charges (APC) of this publication.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

ANN	Artificial Neural Network
ADF	Average Daylight Factor
AES	Advanced Encryption Standard
AF	Automated Fenestration
ANFIS	Adaptive Neuro-Fuzzy Inference Schemes
APPSM	Annual Photosensor Performance Simulation
В	Matrix that represents the lighting offered by p luminaires in q working places
BRDF	Blinn-phong bidirectional Reflectance Distribution Function
EO	Matrix represents the illumination level outside
BSDF	Bidirectional Scattering Distribution function
DA	Daylight Autonomy
DALI	Digital Addressable Lighting Interface
DF	Daylight Factor
DH	Daylight Harvesting
DLI	Daily Light Integral
DFm	Matrix represents the daylight coefficients
ED	Matrix representing the desired illuminance
TMY	Typical Meteorological Year
E _R	Matrix represents the illuminance required from lighting systems
FMRLC	Fuzzy Model Reference Learning Controller
HDR	High Dynamic Image
HOLIGILM	Hollow Light Guide Interior Illumination Method
IGBC	Indian Green Building Council
IMC	Internal Model Control
LEED	Leadership in Energy and Environmental Design
MCU	MicroController Unit
MIMO	Multi-Input Multi-Output
NLARX	Non-Linear Autoregressive
NZEB	Net Zero Energy Buildings
PAR	Plant Aspect Ratio
PCA	Principal Component Analysis
PKI	Public Key Infrastructure
HDR	High Dynamic Range
LEED	Leadership in Energy and Environmental Design
FPGA	Field Programmable Gate Array
DLC	Daylight Linked Control Systems
ASF	Adaptive Solar Façade
FMRLC	Fuzzy Model Reference Learning Controller

LFPM	Liquid filled prismatic louver
UDI	Useful Daylight Illuminance
ASF	Adaptive Solar Façade
CDI	Characteristic Daylight Illuminance
CDImin	Minimum CDI
sCDI	Spatial CDI
PWM	Pulse Width Modulation
SELV	Safety Extra Low Voltage
SF	Scalar Factor
SVM	Support Vector Machine
TDNN	Time Delay Neural Network
UDI	Useful Daylight Illuminance
VLT	Visible Light Transmittance
WFR	Window-to-Floor Ratio
WSN	Wireless Sensor Network
IVR	Immersive Virtual Reality
WWR	Window to Wall Ratios
CS	Circadian Stimulus
POF	Plastic Optical Fiber
WF	Window to Floor Ratios
DF	Daylight Factor
DA	Daylight Autonomy
WEWR	Window-to-Exterior-Wall Ratio
Y	Product of the total lamps (p) and the total number of workspace (q)

References

- 1. Madurai Elavarasan, R.; Pugazhendhi, R.; Jamal, T.; Dyduch, J.; Arif, M.T.; Manoj Kumar, N.; Shafiullah, G.; Chopra, S.S.; Nadarajah, M. Envisioning the UN Sustainable Development Goals (SDGs) through the lens of energy sustainability (SDG 7) in the post-COVID-19 world. *Appl. Energy* **2021**, 116665. [CrossRef]
- Madurai Elavarasan, R.; Selvamanohar, L.; Raju, K.; Rajan Vijayaraghavan, R.; Subburaj, R.; Nurunnabi, M.; Khan, I.A.; Afridhis, S.; Hariharan, A.; Pugazhendhi, R.; et al. A Holistic Review of the Present and Future Drivers of the Renewable Energy Mix in Maharashtra, State of India. Sustainability 2020, 12, 6596. [CrossRef]
- Madurai Elavarasan, R.; Afridhis, S.; Vijayaraghavan, R.R.; Subramaniam, U.; Nurunnabi, M. SWOT analysis: A framework for comprehensive evaluation of drivers and barriers for renewable energy development in significant countries. *Energy Reports* 2020, 6, 1838–1864. [CrossRef]
- 4. Elavarasan, R.M. Comprehensive Review on India's Growth in Renewable Energy Technologies in Comparison With Other Prominent Renewable Energy Based Countries. J. Sol. Energy Eng. 2020, 142. [CrossRef]
- 5. Elavarasan, R.; Shafiullah, G.; Manoj Kumar, N.; Padmanaban, S. A State-of-the-Art Review on the Drive of Renewables in Gujarat, State of India: Present Situation, Barriers and Future Initiatives. *Energies* **2019**, *13*, 40. [CrossRef]
- Elavarasan, R.M.; Shafiullah, G.M.; Padmanaban, S.; Kumar, N.M.; Annam, A.; Vetrichelvan, A.M.; Mihet-Popa, L.; Holm-Nielsen, J.B. A Comprehensive Review on Renewable Energy Development, Challenges, and Policies of Leading Indian States With an International Perspective. *IEEE Access* 2020, *8*, 74432–74457. [CrossRef]
- Kumar, N.M.; Chopra, S.S.; Chand, A.A.; Elavarasan, R.M.; Shafiullah, G.M. Hybrid Renewable Energy Microgrid for a Residential Community: A Techno-Economic and Environmental Perspective in the Context of the SDG7. *Sustainability* 2020, 12, 3944. [CrossRef]
- Elavarasan, R.M.; Pugazhendhi, R.; Shafiullah, G.M.; Irfan, M.; Anvari-Moghaddam, A. A hover view over effectual approaches on pandemic management for sustainable cities—The endowment of prospective technologies with revitalization strategies. *Sustain. Cities Soc.* 2021, 68, 102789. [CrossRef]
- 9. Page, E.R. Daylight Harvesting to Exceed Artificial Light Maximum. U.S. Patent 10,887,970, Application No. 16/411,364, 5 January 2021.
- 10. Pandharipande, A.; Newsham, G.R. Lighting Controls: Evolution and Revolution. *Lighting Res. Technol.* **2018**, *50*, 115–128. [CrossRef]
- 11. Chen, J.J. Daylight Harvest Lighting Control System. U.S. Patent 9,006,982, 24 December 2011.
- 12. Kumar, R. New Algorithms for Daylight Harvesting in a Private Office. In Proceedings of the 2015 18th International Conference on Information Fusion (Fusion) (IEEE), Washington, DC, USA, 6–9 July 2015; pp. 383–392.
- 13. Day, J.K.; Futrell, B.; Cox, R.; Ruiz, S.N.; Amirazar, A.; Zarrabi, A.H.; Azarbayjani, M. Blinded by the Light: Occupant Perceptions and Visual Comfort Assessments of Three Dynamic Daylight Control Systems and Shading Strategies. *Build. Environ.* **2019**, *154*, 107–121.

- Ivanov, D.; Petrinska, I. Lighting Energy Savings through Implementation of Lighting Control System Based on Evolutionary optimisation Algo-rithm. In Proceedings of the 2018 20th International Symposium on Electrical Apparatus and Technologies (SIELA), Bourgas, Bulgaria, 3–6 June 2018.
- Petrinska, I.; Georgiev, V.; Ivanov, D. Lighting Control System for Public Premises, Based on Evolutionary Optimization Algorithm. In Proceedings of the 2018 20th International Symposium on Electrical Apparatus and Technologies (SIELA), Bourgas, Bulgaria, 3–6 June 2018.
- 16. Boscarino, G.; Moallem, M. Daylighting Control and Simulation for LED-Based Energy-Efficient Lighting Systems. *IEEE Trans. Ind. Inform.* **2015**, *12*, 301–309. [CrossRef]
- 17. Mutua, P.W.; Mbuthia, M. Intelligent Multi-Coloured Lighting System Design with Fuzzy Logic Controller. *APTIKOM J. Comput. Sci. Inf. Technol.* **2016**, *1*, 128–140. [CrossRef]
- 18. Liu, J.; Zhang, W.; Chu, X.; Liu, Y. Fuzzy Logic Controller for Energy Savings in a Smart LED Lighting System Considering Lighting Comfort and Daylight. *Energy Build*. **2016**, *127*, 95–104.
- 19. Velasquez, J.J.; Passino, K.M. Fuzzy Fault Tolerant Control for Smart Lights. J. Intell. Fuzzy Syst. 2015, 28, 2605–2620. [CrossRef]
- Colohan, A.; Teehan, J.; Sunderland, K.; Barrett, M.; Preston, J. Digital Energy Networks: A Post Occupancy Evaluation and Appraisal of an Intelligent Low Energy Lighting System. In Proceedings of the 2015 50th International Universities Power Engineering Conference (UPEC), Stoke on Trent, UK, 1–4 September 2015; pp. 1–6.
- Mohagheghi, A.; Moallem, M.; Khayatian, A. Neural Network-Based LED Lighting Control with Modeling Uncertainty and Daylight Disturbance. In Proceedings of the IECON 2017-43rd Annual Conference of the IEEE Industrial Electronics Society, Beijing, China, 29 October–1 November 2017; pp. 3627–3632.
- 22. Attarchi, S.; Moallem, M. Set-Point Control of LED Luminaires for Daylight Harvesting. In Proceedings of the 2017 5th International Conference on Control, Instrumentation, and Automation (ICCIA), Shiraz, Iran, 21–23 November 2017; pp. 244–248.
- Kandasamy, N.K.; Karunagaran, G.; Spanos, C.; Tseng, K.J.; Soong, B.H. 2018. Smart Lighting System Using ANN-IMC for Personalized Lighting Control and Daylight Harvesting. *Build. Environ.* 2018, 139, 170–180. [CrossRef]
- 24. Seyedolhosseini, A.; Masoumi, N.; Modarressi, M.; Karimian, N. Daylight Adaptive Smart Indoor Lighting Control Method Using Artificial Neural Networks. *J. Build. Eng.* **2020**, *29*, 101141. [CrossRef]
- 25. Li, S.; Pandharipande, A.; Willems, F.M. Daylight Sensing LED Lighting System. IEEE Sens. J. 2016, 16, 3216–3223. [CrossRef]
- Gao, Y.; Cheng, Y.; Zhang, H.; Zou, N. Dynamic Illuminance Measurement and Control Used for Smart Lighting with LED. Measurement 2019, 139, 380–386. [CrossRef]
- 27. Wang, Y.; Dasgupta, P. Designing an Adaptive Lighting Control System for Smart Buildings and Homes. In Proceedings of the 2015 IEEE 12th International Conference on Networking, Sensing and Control, Taipei, Taiwan, 9–11 April 2015; pp. 450–455.
- Bouroussis, C.A.; Topalis, F.V. Smart Multi-Workplane Lighting Control and Utilization of Daylight Using an Imaging Photosensor. In Proceedings of the 2016 IEEE 16th International Conference on Environment and Electrical Engineering (EEEIC), Florence, Italy, 7–10 June 2016; pp. 1–6.
- Jiang, J.; Moallem, M. Development of Greenhouse LED System with RedlBlue Mixing Ratio and Daylight Control. In Proceedings of the 2018 IEEE Conference on Control Technology and Applications (CCTA), Copenhagen, Denmark, 21–24 August 2018; pp. 1197–1202.
- 30. Jiang, J.; Mohagheghi, A.; Moallem, M. Energy-Efficient Supplemental LED Lighting Control for a Proof-of-Concept GreenHouse System. *IEEE Trans. Ind. Electron.* 2019, 67, 3033–3042. [CrossRef]
- 31. Tang, S.; Kalavally, V.; Ng, K.Y.; Parkkinen, J. Development of a Prototype Smart Home Intelligent Lighting Control Architecture Using Sensors Onboard a Mobile Computing System. *Energy Build.* **2017**, *138*, 368–376. [CrossRef]
- 32. Zhang, J.; Yin, Z.; Jin, P. Error Analysis and Auto Correction of Hybrid Solar Tracking System Using Photo Sensors and Orientation Algorithm. *Energy* **2019**, *182*, 585–593. [CrossRef]
- Doulos, L.T.; Kontadakis, A.; Madias, E.N.; Sinou, M.; Tsangrassoulis, A. Minimizing Energy Consumption for Artificial Lighting in a Typical Classroom of a Hellenic Public School Aiming for Near Zero Energy Building Using LED DC Luminaires and Daylight Harvesting Systems. *Energy Build.* 2019, 194, 201–217. [CrossRef]
- 34. Li, X.; Wei, Y.; Zhang, J.; Jin, P. Design and Analysis of an Active Daylight Harvesting System for Building. *Renew. Energy* **2019**, 139, 670–678. [CrossRef]
- 35. Bellia, L.; Fragliasso, F. Automated Daylight-Linked Control Systems Performance with Illuminance Sensors for Side-Lit Offices in the Mediterranean Area. *Autom. Constr.* **2019**, *100*, 145–162. [CrossRef]
- 36. Hillmer, H.; Al-Qargholi, B.; Khan, M.M.; Worapattrakul, N.; Wilke, H.; Woidt, C.; Tatzel, A. Optical MEMS-Based Micromirror Arrays for Active Light Steering in Smart Windows. *Jpn. J. Appl. Phys.* **2018**, *57*, 08PA07. [CrossRef]
- Dikel, E.E.; Newsham, G.R.; Xue, H.; Valdés, J.J. Potential Energy Savings from High-Resolution Sensor Controls for LED Lighting. Energy Build. 2018, 158, 43–53. [CrossRef]
- Bellia, L.; Fragliasso, F. Evaluating Performance of Daylight-Linked Building Controls during Preliminary Design. *Autom. Constr.* 2018, 93, 293–314. [CrossRef]
- Omar, O.; García-Fernández, B.; Fernández-Balbuena, A.Á.; Vázquez-Moliní, D. Optimization of Daylight Utilization in Energy Saving Application on the Library in Faculty of Architecture, Design and Built Environment, Beirut Arab University. *Alex. Eng. J.* 2018, 57, 3921–3930. [CrossRef]

- 40. Tabadkani, A.; Shoubi, M.V.; Soflaei, F.; Banihashemi, S. Integrated Parametric Design of Adaptive Facades for User's Visual Comfort. *Autom. Constr.* 2019, 106, 102857. [CrossRef]
- 41. Ohki, C.; Okamoto, T.; Ohga, H.; Yoshizawa, N. Energy Performance Evaluation of Outside Sun Shadings Using Radiance and Newhasp. J. Environ. Eng. Jpn. 2018, 83, 861–870. [CrossRef]
- 42. Wu, Y.; Kämpf, J.H.; Scartezzini, J.L. Design and validation of a compact embedded photometric device for real-time daylighting computing in office buildings. *Build. Environ.* **2019**, *148*, 309–322. [CrossRef]
- Kunwar, N.; Cetin, K.S.; Passe, U.; Zhou, X.; Li, Y. Full-Scale Experimental Testing of Integrated Dynamically-Operated Roller Shades and Lighting in Perimeter Office Spaces. Sol. Energy 2019, 186, 17–28. [CrossRef]
- 44. Verma, N.; Jain, A. Optimized Automatic Lighting Control in a Hotel Building for Energy Efficiency. In Proceedings of the 2018 International Conference on Power Energy, Environment and Intelligent Control (PEEIC), Greater Noida, India, 13–14 April 2018; pp. 168–172.
- 45. Alva, M.; Vlachokostas, A.; Madamopoulos, N. Experimental Demonstration and Performance Evaluation of a Complex Fenestration System for Daylighting and Thermal Harvesting. *Sol. Energy* **2020**, *197*, 385–395. [CrossRef]
- 46. Vlachokostas, A.; Madamopoulos, N. Liquid Filled Prismatic Louver Façade for Enhanced Daylighting in High-Rise Commercial Buildings. *Opt. Express* **2015**, *23*, A805–A818. [CrossRef]
- 47. Samadi, S.; Noorzai, E.; Beltrán, L.O.; Abbasi, S. A Computational Approach for Achieving Optimum Daylight inside Buildings through Automated Kinetic Shading Systems. *Front. Archit. Res.* **2020**, *9*, 335–349. [CrossRef]
- Zhang, Z.; Ding, H.; Alem, S.; Griffin, R.; Tao, Y.; Xiao, G.; Pardasani, A.; Xu, H.; Arsenault, C.; Newsham, G. Printed Sensors for Building Management. In Proceedings of the 2018 International Flexible Electronics Technology Conference (IFETC), Ottawa, ON, Canada, 7–9 August 2018; pp. 1–3.
- 49. Sun, X.; Gou, Z.; Lau, S.S.Y. Cost-Effectiveness of Active and Passive Design Strategies for Existing Building Retrofits in Tropical Climate: Case Study of a Zero Energy Building. *J. Clean. Prod.* **2018**, *183*, 35–45. [CrossRef]
- 50. McCunn, L.J.; Kim, A.; Feracor, J. Reflections on a Retrofit: Organizational Commitment, Perceived Productivity and Controllability in a Building Lighting Project in the United States. *Energy Res. Soc. Sci.* **2018**, *38*, 154–164. [CrossRef]
- 51. Satumane, A.D.; Suk, J.Y. Impacts of Vertical Greening System (VGS) on Daylight Quantity and Quality in Buildings. In Proceedings of the ARCC Conference Repository, Pennsylvania, PHL, USA, 25 September 2018.
- Wang, X.; Tjalkens, T.; Linnartz, J.P. Smart Office Lighting Control Using Occupancy Sensors. In Proceedings of the 2017 IEEE 14th International Conference on Networking, Sensing and Control (ICNSC), Calabria, Italy, 16–18 May 2017; pp. 453–458.
- 53. Zehra, F.; Korkmaz, K.A.; Ahmeda, M.S. Sustainable Lighting System for University Buildings. J. Build. Sustain. 2018, 1, 12–22.
- Özçelik, M.A. The Design and Comparison of Central and Distributed Light Sensored Smart LED Lighting Systems. *Int. J. Photoenergy.* 2018, 2018, 1–14. [CrossRef]
- Sripramong, S.; Ananwattanaporn, S.; Ngaopitakkul, A.; Jettanasen, C. A Study an Impact of Daylight with Period Time on Illuminance and Uniformity for Building. In Proceedings of the International MultiConference of Engineers and Computer Scientists, Hongkong, China, 18–20 March 2015; Volume 2.
- Maskarenj, M.S.; Avasare, M.; Ghosh, P.C. Analysis of Plastic Optical Fiber Based Daylight System Suitable for Building Applications. In *Applied Mechanics and Materials*; Trans Tech Publications Ltd.: Schwyz, Switzerland, 2014; Volume 492, pp. 101–105.
- Ashdown, I.; Scott, W.J. System and Method for the Optimization of Radiance Modeling and Controls in Predictive Daylight Harvesting. U.S. Patent Application No. 15/407,176, 14 May 2019.
- 58. Hu, J.; Olbina, S. Simulation-Based Model for Integrated Daylighting System Design. J. Comput. Civ. Eng. 2014, 28, A4014003. [CrossRef]
- 59. Yoon, Y.; Lee, J.H.; Kim, S. Development of Computational Algorithm for Prediction of Photosensor Signals in Daylight Conditions. *Build. Environ.* **2015**, *89*, 229–243. [CrossRef]
- 60. Bellia, L.; Fragliasso, F.; Pedace, A. Evaluation of Daylight Availability for Energy Savings. J. Daylighting 2015, 2, 12–20. [CrossRef]
- 61. Shen, E.; Hu, J.; Patel, M. Energy and Visual Comfort Analysis of Lighting and Daylight Control Strategies. *Build. Environ.* 2014, 78, 155–170. [CrossRef]
- 62. Devaraj, J.; Madurai Elavarasan, R.; Shafiullah, G.; Jamal, T.; Khan, I. A holistic review on energy forecasting using big data and deep learning models. *Int. J. Energy Res.* 2021, er.6679. [CrossRef]
- 63. Zhou, S.; Liu, D. Prediction of Daylighting and Energy Performance Using Artificial Neural Network and Support Vector Machine. *Am. J. Civ. Eng. Archit.* **2015**, *3*, 1–8.
- 64. Navada, S.G.; Adiga, C.S.; Kini, S.G. Prediction of Daylight Availability for Visual Comfort. Int. J. Appl. Eng. Res. 2016, 11, 4711–4717.
- 65. Yi, Y.K. Dynamic Coupling between a Kriging-Based Daylight Model and Building Energy Model. *Energy Build.* **2016**, *128*, 798–808. [CrossRef]
- 66. Ashdown, I. Analytic BSDF Modeling for Daylight Design. Available online: https://www.researchgate.net/profile/ Ian-Ashdown/publication/326901421_Analytic_BSDF_Modeling_for_Daylight_Design/links/5b6af6a992851ca650515240 /Analytic-BSDF-Modeling-for-Daylight-Design.pdf. (accessed on 2 January 2021).
- 67. Kómar, L.; Kocifaj, M. An Accurate Prediction of Daylight Pipe Harvesting of Interior Space. Appl. Sci. 2019, 9, 3552. [CrossRef]

- 68. Kocifaj, M.; Petržala, J. Designing of Light-Pipe Diffuser through Its Computed Optical Properties: A Novel Solution Technique and some Consequences. *Sol. Energy* **2019**, *190*, 386–395. [CrossRef]
- 69. Alves, T.; Machado, L.; de Souza, R.G.; de Wilde, P. Assessing the Energy Saving Potential of an Existing High-Rise Office Building Stock. *Energy Build.* 2018, 173, 547–561. [CrossRef]
- 70. Humann, C. Expert System for Prediction of Changes to Local Environment. U.S. Patent 9,406,028, 2 August 2016.
- Colaco, S.G.; Colaco, A.M.; Kurian, C.P.; George, V.I. An Adaptive Predictive Framework to Online Prediction of Interior Daylight Illuminance. In Proceedings of the 2014 International Conference on Advances in Energy Conversion Technologies (ICAECT), Manipal, India, 23–25 January 2014; pp. 174–180.
- 72. Harris, B.; Montes, J.; Forbes, N. A Novel Approach to Sensor-Less Daylight Harvesting in Commercial Office Buildings. In *Asset Intelligence through Integration and Interoperability and Contemporary Vibration Engineering Technologies*; Springer: Cham, Switzerland, 2019; pp. 209–218.
- 73. Awad, J.; Abd-Rabo, L. Daylight and Energy Performance Optimization in Hot-Arid Regions: Application and Adaptation Guide for Designers in the UAE. *Procedia Manuf.* 2020, 44, 237–244. [CrossRef]
- 74. Wang, J.; Wei, M.; Chen, L. Does Typical Weather Data Allow Accurate Predictions of Daylight Quality and Daylight-Responsive Control System Performance. *Energy Build.* **2019**, *184*, 72–87. [CrossRef]
- Mathew, M.; Kini, S.G. Daylight Analysis of a Typical Low Rise Office Building for Different Climate Zones of India. In Proceedings of the 2016 International Conference on Energy Efficient Technologies for Sustainability (ICEETS), Nagercoil, India, 7–8 April 2016; pp. 549–553.
- 76. Mohapatra, B.N.; Kumar, M.R.; Mandal, S.K. Analysis of Daylighting Using Daylight Factor and Luminance for Different Room Scenarios. *Int. J. Civ. Eng. Technol. IJCIET* **2018**, *9*, 949–960.
- 77. Nasrollahi, N.; Shokry, E. Parametric Analysis of Architectural Elements on Daylight, Visual Comfort, and Electrical Energy Performance in the Study Spaces. *J. Daylighting* **2020**, *7*, 57–72. [CrossRef]
- deRubeis, T.; Nardi, I.; Muttillo, M.; Ranieri, S.; Ambrosini, D. Room and Window Geometry Influence for Daylight Harvesting Maximization–Effects on Energy Savings in an Academic Classroom. *Energy Procedia* 2018, 148, 1090–1097.
- Yoshizawa, N.; Miki, Y.; Yamaguchi, H.; Tamura, H. Systematic Evaluation on Daylight Harvesting Energy-Saving Effects in Offices: Through Daylighting Simulation Using Annual Standard Weather Data in Tokyo. J. Environ. Eng. Jpn. 2018, 83, 425–434. [CrossRef]
- 80. Ghasemi, M.; Kandar, M.Z.; Noroozi, M. Investigating the Effect of Well Geometry on the Daylight Performance in the Adjoining Spaces of Vertical Top-Lit Atrium Buildings. *Indoor Built Environ.* **2016**, *25*, 934–948. [CrossRef]
- 81. Ashrafian, T.; Moazzen, N. The Impact of Glazing Ratio and Window Configuration on Occupants' Comfort and Energy Demand: The Case Study of a School Building in Eskisehir, Turkey. *Sustain. Cities Soc.* **2019**, *47*, 101483. [CrossRef]
- 82. Yaw-Shyan, T.S.A.Y.; Chih-Hung, Y.A.N.G. The Influence on Daylight and Energy Consumption of Expanded Metal Mesh Applied on Building Façades. In Proceedings of the E3S Web of Conferences, Bucharest, Romania, 26–29 May 2019; Volume 111.
- 83. Svirsky, A. Adjustable All-Season Window Awning/Light Shelf and Operating Mechanism Therefore. U.S. Patent Application No. 13/007,980, 7 June 2016.
- 84. Gutiérrez, R.U.; Du, J.; Ferreira, N.; Ferrero, A.; Sharples, S. Daylight Control and Performance in Office Buildings Using a Novel Ceramic Louvre System. *Build. Environ.* **2019**, 151, 54–74. [CrossRef]
- 85. Monteoliva, J.M.; Bellia, L.; Fragliasso, F.; Pattini, A. Ancient Romans and Daylighting: The Case of Villa of the Mysteries in Pompeii. *J. Cult. Herit.* **2020**, *43*, 204–218. [CrossRef]
- 86. Amirkhani, M.; Garcia-Hansen, V.; Isoardi, G.; Allan, A. Innovative Window Design Strategy to Reduce Negative Lighting Interventions in Office Buildings. *Energy Build.* **2018**, *179*, 253–263. [CrossRef]
- 87. Acosta, I.; Campano, M.Á.; Leslie, R.; Radetsky, L. Daylighting Design for Healthy Environments: Analysis of Educational Spaces for Optimal Circadian Stimulus. *Sol. Energy* **2019**, *193*, 584–596. [CrossRef]
- 88. Vasilakopoulou, K.; Kolokotsa, D.; Santamouris, M.; Kousis, I.; Asproulias, H.; Gi-annarakis, I. Analysis of the Experimental Performance of Light Pipes. *Energy Build.* **2017**, *151*, 242–249. [CrossRef]
- 89. Shao, L.; Elmualim, A.A.; Yohannes, I. Mirror Lightpipes: Daylighting Performance in Real Buildings, Light. *Res. Technol.* **1998**, 30, 37–44. [CrossRef]
- Baroncini, C.; Chella, F.; Zazzini, P. Experimental Analysis of Tubular Light Pipes Performances: Influence of the Diffuser on Inside Distribution of Light. In Proceedings of the 5th International Conference on Sustainable Energy Technologies, SET, Vicenza, Italy, 30 August–1 September 2006.
- 91. Yanpeng, W.; Chongfang, M. Daylight Performance of Top Lighting Light Pipes and Side Lighting Light. In Proceedings of the World Sustainable Building Conference, Tokyo, Japan, 27–29 September 2005.
- Shailesh, K.R.; Raikar, T.S. Computational Analysis of Daylight Harvesting Scheme in an Office Building in Mumbai. In Proceedings of the 2010 IEEE International Conference on Sustainable Energy Technologies (ICSET), Kandy, Sri Lanka, 6–9 December 2010; pp. 1–8.
- 93. Nezamdoost, A.; Van Den Wymelenberg, K.; Mahic, A. Assessing the Energy and Daylighting Impacts of Human Behavior with Window Shades, a Life-Cycle Comparison of Manual and Automated Blinds. *Autom. Constr.* **2018**, *92*, 133–150. [CrossRef]